Author's response to reviews

Title: Topically applied vitamin C increases the density of dermal papillae in aged human skin

Authors:

Kirsten Sauermann (kirsten.sauermann@beiersdorf.com)
Mr Soren Jaspers (soeren.jaspers@beiersdorf.com)
Urte Koop (urte.koop@beiersdorf.com)
Horst Wenck (horst.wenck@beiersdorf.com)

Version: 2 Date: 2 May 2004

Suggested modifications concerning the references have been made.